Author’s response to reviews

Title: Promoting state health department evidence-based cancer and chronic disease prevention programs: A multi-phase dissemination study with a cluster randomized trial component

Authors:

Peg Allen (pallen@brownschool.wustl.edu)
Sonia Sequeira (soniabsequeira@gmail.com)
Rebekah R Jacob (rjacob2@brownschool.wustl.edu)
Adriano Akira Hino (akira_manaca@yahoo.com.br)
Katherine A Stamatakis (kstamata@slu.edu)
Jenine K Harris (jharris@brownschool.wustl.edu)
Lindsay Elliott (lindsayelliott@brownschool.wustl.edu)
Jon F Kerner (jon.kerner@partnershipagainstcancer.ca)
Ellen Jones (elljax@aol.com)
Maureen Dobbins (dobbinsm@mcmaster.ca)
Elizabeth A Baker (bakerpa@slu.edu)
Ross C Brownson (rbrownson@wustl.edu)

Version: 3
Date: 6 December 2013

Author’s response to reviews: see over
December 6, 2013

TO: Implementation Science Editorial Team

RE: MS 8733243601101355

This is a post-copyedit resubmission of manuscript MS 8733243601101355 to Implementation Science in the “study protocol” category originally entitled “A multi-phase dissemination study: Promoting uptake of evidence-based cancer and chronic disease prevention programs.”

I have checked the copy edits, as did one coauthor. I accepted all of the copy editor’s track changes. I made no content changes. The copy editor did not have any questions for us.

We are prepared to pay the article processing fee at the appropriate time. Washington University is a supporting member of BioMed Central. Proof of our funding award from the National Cancer Institute and human subjects approval was submitted in the original submission. The authors do not have any competing interests. All authors have read and agreed to the manuscript content.

If you have any questions, please contact me at pallen@brownschool.wustl.edu, 314-935-0116.

Thank you,

Peg Allen, PhD
Assistant Research Professor
Prevention Research Center in St. Louis
Washington University in St. Louis